Author's response to reviews

Title: Mitral Annular Calcification Predicts Immediate Results of Percutaneous Transvenous Mitral Commissurotomy

Authors:

Mojtaba Salarifar (sali1959@yahoo.com)
Mehrnaz Rezvanfard (mehrnaz.rezvanfard@gmail.com)
Hakimeh Sadeghian (sadeghianhakimeh@yahoo.com)
Azam Safir-mardanloo (Drsafir.thc@gmail.com)
Nahid Shafii (Shafieenhd@yahoo.com)

Version: 4 Date: 12 October 2011

Author's response to reviews: see over
To The Journal of Ultrasound Echocardiography Independent Editorial Production Team:

Attached please find the second revised version of manuscript entitled "Mitral Annular Calcification Predicts Immediate Results of Percutaneous Transvenous Mitral Commissurotomy" which is edited for consideration of editors of "Journal of Ultrasound Cardiology". The manuscript is modified according to the requested changes to the format of the manuscript. We edited the figure legends and additional files' descriptions and table format according to the journal style and also made changes according to the minor revisions. We Hope that the current version of the manuscript is now in accordance with the journal style.

Sincerely

Hakimeh Sadeghian

Associate professor of Cardiology

Octbre 12th/2011